Author’s response to reviews

Title: Comparative efficacy of Chinese medicine injections for treating acute cerebral infarction: A network meta-analysis of randomized controlled trials

Authors:

Shi Liu (1048823645@qq.com)
Jia-Rui Wu (exogamy@163.com)
Dan Zhang (2426394372@qq.com)
Kai-Huan Wang (wkhlinda@163.com)
Bing Zhang (zhangbing6@263.net)
Xiao-Meng Zhang (fensexiaomao@sina.com)
Di Tan (tandi1993@qq.com)
Xiao-Jiao Duan (duanxiaojiaodgh@163.com)
Ying-Ying Cui (867422664@qq.com)
Xin-Kui Liu (1020172827@qq.com)

Version: 1 Date: 01 Dec 2017

Author’s response to reviews:

Dear editor,

I am very grateful to your comments for the manuscript. According with your advice, we amended the relevant part in manuscript. Some of your questions were answered below.

1. Please include the email addresses of all authors on the title page of your manuscript.

The authors' answer: Thanks for the comments.

We have listed email addresses of all authors on the title page of manuscript.

2. Please provide a Keywords section after your Abstract that includes three to ten keywords representing the main content of the article.
The authors’ answer: Thanks for the comments.

We have added Keywords section.

3. Please revise your Competing Interests section to account for all possible conflicts, not just financial ones.

The authors’ answer: Thanks for the comments.

There was not any competing interests between authors, and we had add statement in the section titled “Competing interests”

4. Thank you for providing a section on author contribution. We use this information to note which author contributed what to the study. We had noticed only nine of the ten authors are mentioned in this section. Please revise this section to include all authors, so that all authors listed in the manuscript can qualify for authorship.

An 'author' is generally considered to be someone who has made substantive intellectual contributions to a published study. We respectfully request you please amend this statement to adhere to the guidelines for authorship that are applicable in your research field or, in the absence of any guidelines, to the International Committee of Medical Journal Editors (ICMJE) guidelines. According to the ICMJE guidelines, to qualify as an author one should have:

- made substantial contributions to conception and design, or acquisition of data, or analysis and interpretation of data;

- been involved in drafting the manuscript or revising it critically for important intellectual content;

- given final approval of the version to be published. Each author should have participated sufficiently in the work to take public responsibility for appropriate portions of the content; and

- agreed to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved.

Acquisition of funding, collection of data, or general supervision of the research group, alone, does not usually justify authorship.

If you would please amend the statement to reflex these requirements, we would appreciate it.

Additionally, please use this statement to confirm that all authors have read and approved of the final manuscript.

The authors’ answer: Thanks for the comments.
We had amended “Author contributions” conformed to our research in the section titled “Author contributions”.

4. Please do not include the figure captions within the figure files, but instead place them at the end of the main manuscript in a section titled “Figure legends”.

The authors’ answer: Thanks for the comments.

We had placed the note in the section titled “Figure legends”.

Thank you for your comments again.

Sincerely,

Jia-Rui Wu

Department of Clinical Chinese Pharmacy
School of Chinese Materia Medica
Beijing University of Chinese Medicine
Beijing, China
Tel: 010-6428-7506
E-mail: exogamy@163.com